Author's response to reviews

Title: MRCP compared to diagnostic ERCP for the diagnosis of biliary obstruction: a systematic review

Authors:

Eva C Kaltenthaler (e.kaltenthaler@sheffield.ac.uk)
Stephen J Walters (s.j.walters@sheffield.ac.uk)
Jim B Chilcott (j.b.chilcott@sheffield.ac.uk)
Anthony Blakeborough (Tony.Blakeborough@sth.nhs.uk)
Yolanda Bravo Vergel (yb3@york.ac.uk)
Steven Thomas (s.m.thomas@sheffield.ac.uk)

Version: 4  Date: 24 July 2006

Author's response to reviews:

The Editor
BMC Medical Imaging

24 July 2006

Journal BMC Medical Imaging
MS: 5832265421015545

Dear Editors,

Please find attached an amended version of our manuscript "MRCP compared to diagnostic ERCP for the diagnosis of biliary obstruction: a systematic review."

We have made the following formatting changes as requested:
1. We have added the country (United Kingdom) to the authors’ details on the title page.
2. We have removed the table and figure placement text within the manuscript.
3. We have changes the List of abbreviations to Abbreviations.
4. We have amended all references to include all authors and removed "et al."
5. The image files no longer include the figure or table title and number. A Figure legend has been created after the references and all figures and tables are listed there in the order in which they are referred to in the text.
6. All files and tables are refereed to in the text with the term "see Additional file."

Thank you for accepting this article for publication in BMC Medical Imaging.

Yours faithfully,

Dr Eva Kaltenthaler
Senior Research Fellow
ScHARR
University of Sheffield